Author's response to reviews

Title: Positive outcome of average volume-assured pressure support mode of a Respironics V60 Ventilator in acute exacerbation of Chronic Obstructive Pulmonary Disease: a case report.

Authors:

MIYUKI OKUDA (ryoheikunn@msf.biglobe.ne.jp)
MAKOTO KASHIO (ryoheikunn@msf.biglobe.ne.jp)
NOBUYA TANAKA (ryoheikunn@msf.biglobe.ne.jp)
TAKASHI FUJII (ryoheikunn@msf.biglobe.ne.jp)
YOSHINARI OKUDA (ryoheikunn@msf.biglobe.ne.jp)

Version: 7 Date: 30 March 2012

Author's response to reviews: see over
Dear The Journal of medical case reports Editorial Team

Thank you for your reviewing our manuscript “Positive outcome of average volume-assured pressure support mode of a Respironics V60 Ventilator in acute exacerbation of Chronic Obstructive Pulmonary Disease: a case report.”

We would answer the reviewers’ comments as following.

Please add the trade mark symbol to the usages of the phrase “Respironics V60 Ventilator”.

Respironics V60 Ventilator is not registered as a trademark

We advise that you change the title of your manuscript to:
“Positive outcome of average volume-assured pressure support mode of a Respironics V60 Ventilator in acute exacerbation of Chronic Obstructive Pulmonary Disease: a case report.

We agree with your comment.

We rewrote it according to your opinion.

Thank you for your comments and queries regarding this manuscript.
We would like to respectfully resubmit it for your consideration and hopeful acceptance.
Please let me know if you have any problems or questions in the meantime.

Sincerely,
Miyuki Okuda